Author's response to reviews

Title: Significantly higher serum concentration of alpha-1 antitrypsin in colorectal cancer patients than in healthy controls

Authors:

Sergio Perez-Holanda (perezholandas@gmail.com)
Ignacio Blanco (ignablanco@yahoo.com)
Manuel Menéndez (perezholandas@gmail.com)
Luis Rodrigo (lrodrigosaez@gmail.com)

Version: 5 Date: 10 April 2014

Author's response to reviews:

Dear Editor:

We would like to first thank the Editorial Board and the Referees for their interest in the evaluation of our manuscript, as well as for their valuable suggestions and useful ideas to improve the quality of our paper.

Please find below the issues raised by both reviewers, and our point by point responses.